Author's response to reviews

Title: Acute bilateral anterior circulation stroke due to anomalous cerebral vasculature: a case report

Authors:

Brian F Menezes (franmenezes77@yahoo.co.uk)
Beverly J Cheserem (bevjeb@yahoo.co.uk)
Jothy Kandasamy (Jothy.Kandasamy@thewaltoncentre.nhs.uk)

Version: 3 Date: 27 March 2008

Author's response to reviews:

Dear Editor,
Following further suggestions from the reviewers, I would like to submit a revision of my case report.
A response to the reviewers' suggestions is enclosed herein.
RESPONSE TO REVIEWERS

Reviewer: DEBORAH LOWE
Thank you for the interest shown, for the helpful comments and for the support in approving the acceptance of our case report.

Reviewer: DAVID TONG
T1/T2 weighted MR images supporting the initial suspicion of a bifrontal space occupying lesion have now been attached in this revised version of the case report. Unfortunately the first MR images done less than 8 hours after the initial presentation (and which did not show areas of high signal) were performed in another Trust and have since been irretrievable (due to misplacement). We apologise for this deficiency and would like to thank you for your invaluable suggestions and recommendations.